Author's response to reviews

Title: Evidence for a Gene Influencing Heart Rate on Chromosome 5p13-14 in a Meta-Analysis of Genome-Wide Scans from the NHLBI Family Blood Pressure Program

Authors:

Jason M Laramie (laramiej@bu.edu)
Jemma B Wilk (jwilk@bu.edu)
Steven C Hunt (steve@sch.med.utah.edu)
Curtis Ellison (ellison@bu.edu)
Aravinda Chakravarti (aravinda@jhmi.edu)
Eric Boerwinkle (Eric.Boerwinkle@uth.tmc.edu)
Richard H Myers (rmyers@bu.edu)

Version: 5 Date: 31 January 2006

Author's response to reviews:

Added the sentence in the Methods section under the subheading Study Participants - "Each network has been approved by the appropriate local Institutional Review Board and all participants provided informed written consent."

Added two references for HEGESMA under the Methods section under the subheading Statistical Methods.

Moved the last paragraph in the Discussion to the Conclusion Section

Reformatted Table 2 and 3 to fit within standard paper margins.

Added an Additional Files Section.

Removed Authors' qualifications from the title page.